

Format for ANSWERING REVIEWERS

March 13, 2024



Dear Editors,

On behalf of all the authors, I would like to thank you for your consideration of this paper. In the revised manuscript you will find the changes that we made in response to the Reviewers. In this response to reviewer letter we also indicated how we have dealt with the Reviewers' comments.

Please find enclosed the edited manuscript in Word format.

Name of Journal: *World Journal of Gastroenterology*

Manuscript Type: Editorial

SIRT segmentectomy. A new minimally invasive curative option for primary liver malignancies?

Riccardo Inchingolo, Francesco Cortese, Pisani Antonio Rosario, Fabrizio Acquafredda, Roberto Calbi, Riccardo Memeo, Fotis Anagnostopoulos, Stavros Spiliopoulos

Invited Manuscript: 03358964

Manuscript ID.: 91322

The manuscript has been improved according to the suggestions of reviewer and Editorial Office's

Comments:

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Accept (General priority)

Specific Comments to Authors:

In this manuscript authors have presented transarterial radioembolization as curative intent for the treatment of liver cancer. All aspects including properties, mechanism, rationale and future approach presented very well.

Authors' reply: Thank you for the consideration of this paper and for your comment. We did a further English revision with language polishing.

(1) Science editor:

1 Conflict of interest statement: Academic Editor has no conflict of interest.

2 Scientific quality: The author submitted an editorial of transarterial radioembolization is a percutaneous, local, intra-arterial, treatment of cancer performed by injecting radioactive microspheres within the tumour. The manuscript is overall qualified.

Authors' reply: Thank you for the consideration of this paper and for your comment

(1) Advantages and disadvantages: The reviewer have given positive peer-review reports for the manuscript. Classification: Grade B; Language Quality: Grade B. In this manuscript authors have presented transarterial radioembolization as curative intent for the treatment of liver cancer. All aspects including properties, mechanism, rationale and future approach presented very well.

(2) Main manuscript content: The author clearly stated the purpose of the study and the research structure is complete. However, the manuscript is still required a further revision according to the detailed comments listed below.

(3) Table(s) and figure(s): There are 2 Figures should be improved. Detailed suggestions for each are listed in the specific comments section.

(4) References: A total of 50 references are cited, including 29 published in the last 3 years. The reviewer didn't request the authors to cite improper references published by him/herself.

3 Language evaluation: The English-language grammatical presentation needs to be improved to a certain extent. There are many errors in grammar and format, throughout the entire manuscript. Before final acceptance, the authors must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: <https://www.wjgnet.com/bpg/gerinfo/240>.

Authors' reply: Thank you for the consideration of this paper and for your comment. A native speaker English did a further English revision with language polishing.

4 Medical ethics: Please provide the filled conflict-of-interest disclosure form.

5 Specific comments: (1) Please provide the Figures cited in the original manuscript in the form of PPT. All text can be edited, including A, B, arrows, etc. All legends are incorrectly formatted and require a general title and explanation for each figure. Such as Figure 1 title. A: ; B: ; C: .

Authors' reply: Thank you for the consideration of this paper and for your comment. We did the requested changes.

(2) Please obtain permission for the use of picture(s). If an author of a submission is re-using a figure or figures published elsewhere, or that is copyrighted, the author must provide documentation that the previous publisher or copyright holder has given permission for the figure to be re-published, and correctly indicate the reference source and copyrights. For example, "**Figure 1 Histopathological examination by hematoxylin-eosin staining (200 x)**. A: Control group; B: Model group; C: Pioglitazone hydrochloride group; D: Chinese herbal medicine

group. **Citation:** Yang JM, Sun Y, Wang M, Zhang XL, Zhang SJ, Gao YS, Chen L, Wu MY, Zhou L, Zhou YM, Wang Y, Zheng FJ, Li YH. Regulatory effect of a Chinese herbal medicine formula on non-alcoholic fatty liver disease. *World J Gastroenterol* 2019; 25(34): 5105-5119. Copyright ©The Author(s) 2019. Published by Baishideng Publishing Group Inc^[6]. And please cite the reference source in the references list. If the author fails to properly cite the published or copyrighted picture(s) or table(s) as described above, he/she will be subject to withdrawal of the article from BPG publications and may even be held liable.

6 Recommendation: Conditional acceptance.

Language Quality: Grade B (Minor language polishing)

Scientific Quality: Grade B (Very good)

(2) *Company editor-in-chief:*

I have reviewed the Peer-Review Report, full text of the manuscript, all of which have met the basic publishing requirements of the *World Journal of Gastroenterology*, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office's comments and the Criteria for Manuscript Revision by Authors.

Authors' reply: Thank you for the consideration of this paper and for your comment.

Finally, we wish to thank the Editors and the Reviewer for their comments that helped us to increase the value of our paper.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Riccardo Inchingolo, MD, EBIR, CIRSE Fellow

Interventional Radiology Unit,

"F. Miulli" Regional General Hospital,

Acquaviva delle Fonti (BA), 70021, Italy

riccardoin@hotmail.it

Tel: +39-333-4601735

Fax: +39-0835-253857